Reviewer Comments

Point 1. The authors present the 10 cases over a 23yr period where renal biopsies of congenital heart disease patients were available. To let readers have an understanding of the magnitude of the problem, they should elaborate in the Results or Discussion how many cases of congenital heart disease they handled in this period, and how many cases of both congenital heart and renal disease patients they saw over the similar period. Answer: We added to the discussion the number of patients who are registered at the institute with congenital heart disease, as well as the prevalence that exists.

Point 2. In Table 2, for the last 41yr old patient, sex of "hombre" is inappropriate. Answer: We change the term "hombre" for man in table 2.

Point 3. In their Discussion, the authors should make concrete practical recommendations on when renal biopsy should be performed in patients with both congenital heart and kidney disease. Answer: We added in the methods the indication that these patients had for renal biopsy.

In addition to the corrections made by the reviewer, we made other corrections that we observed, which highlighted contents with yellow color in the revised manuscript.